Author’s response to reviews

Title: Developing content for national population health surveys: an example using a newly developed sedentary behaviour module

Authors:

Stephanie Prince (stephanie.princeware@canada.ca; sprin063@uottawa.ca; s.prince.ware@gmail.com)

Gregory Butler (Gregory.Butler@canada.ca)

Karen Roberts (KarenC.Roberts@canada.ca)

Pam Lapointe (Pam.Lapointe@canada.ca)

Andrew MacKenzie (Andrew.Mackenzie@canada.ca)

Rachel Colley (Rachel.Colley@canada.ca)

Maria Foley (Maria.Foley@canada.ca)

Travis Saunders (trsaunders@upei.ca)

Wendy Thompson (Wendy.Thompson@canada.ca)

Version: 3 Date: 17 Nov 2019

Author’s response to reviews:

Dear Dr. Yokota,

Reference: Developing content for national population health surveys: an example using a newly developed sedentary behavior module (AOPH-D-19-00147R1)

Thank you once again for your careful consideration of our manuscript. We received your request for Minor Revision.

We have now added a legend for Figure 1 under the appropriate figure legend in the Word document. We hope that these revisions are acceptable and address the concerns that were raised.